Dear Dr

Lian-Sheng Ma, Founder and CEO
Lian-Sheng Ma, Science Editor, Company Editor-in-Chief, Editorial Office

Baishideng Publishing Group Inc

We would like to thank the editor and the reviewer for their helpful comments and insight regarding our previous submission. According to the Referee's comments, no questions or considerations were presented regarding the Review. Additionally, Language quality, Grade A: Priority publishing.

Patricia Castelucci, PhD
Associate Professor
University of São Paulo

PEER-REVIEW REPORT

Name of journal: World Journal of Gastroenterology
Manuscript NO: 69749
Title: Enteric nervous system and inflammatory bowel diseases: Correlated impacts and therapeutic approaches through the X7 receptor
Reviewer's code: 05121403
Position: Peer Reviewer
Academic degree: PhD
Despite the great advancement have been made in the field of IBDs, the therapeutic approaches are still limited. The P2X7 receptor has become a promising therapeutic target for IBDs. This review summarizes the impacts caused by IBDs on enteric nervous system cells and brings together the findings of the most recent literature on therapeutic approaches through the P2X7 receptor.